Author's response to reviews

Title: Association of an NFKB1 intron SNP (rs4648068) with gastric cancer patients in the Han Chinese population

Authors:
renquan lu (lurenquan66@hotmail.com)
Xiang gao (gncy2001@hotmail.com)
Yin Chen (1251487447@qq.com)
Jian Ni (jiannihome@yahoo.com)
Yongfu Yu (dearfisher@gmail.com)
Sheng Li (lisheng1996@gmail.com)
Lin Guo (jykca@sina.com)

Version: 7 Date: 12 June 2012

Author's response to reviews: see over
Dear Mr. Editor,

Thank you so much for your kind consideration on our manuscript. According to the suggestions of the reviewer, we have revised the abstract of our manuscript point by point as follows:

The sentence "Furthermore, stratified by lymph node status and serosa invasion, the effect of the rs4648068 GG genotype was larger in gastric cancer patients (AORs.....)." was replaced with "Furthermore, among the gastric cancer cases, the rs4648068 GG genotype was associated with high clinical stage (AOR = 2.27, 95% CI: 1.11- 4.66), lymph node involvement (AOR = 2.90, 95% CI = 1.40 - 6.03) and serosa invasion (AOR = 2.78, 95% CI = 1.34 - 5.75)."

Thank you very much.

Best Regards,

Lin Guo